Author’s response to reviews

Title: Full-term abdominal extrauterine pregnancy complicated by post-operative ascites with successful outcome: a case report

Authors:

Gwinyai Masukume (parturitions@gmail.com)
Elton Sengurayi (esengus@yahoo.com)
Alfred Muchara (atm4jc@gmail.com)
Emmanuel Mucheni (emucheni@yahoo.com)
Wedu Ndebele (wndebele@yahoo.co.uk)
Solwayo Ngwenya (drsolngwe@yahoo.co.uk)

Version: 6 Date: 16 October 2012

Author’s response to reviews:

Dear JMCR,

We were fortunate to acquire additional data about 6 months after the patient’s initial presentation, which we are the opinion is of significant importance. We add this data to our manuscript.

We also made the following relatively minor changes:

1) In the abstract changed ‘had ascites’ to ‘developed ascites’
2) In the table, specified that hemorrhagic shock occurred intra-operatively and postulated the likely cause of the thrombocytopenia
3) Changed pre-renal failure to pre-renal acute kidney injury in the table’s legend
4) Changed intrabdominal to intra-abdominal
5) Mentioned that SN was the primary physician responsible for the patient’s care

Sincerely yours,
GM, ES, AM, EM, WN & SN